Author's response to reviews

Title: Trends in influenza vaccination coverage in Portugal from 1998 to 2010

Authors:

cátia sousa pinto (catiasousapinto@gmail.com)
baltazar Nunes (balta.nunes@insa.min-saude.com)
maria joão branco (m.joao.branco@insa.min-saude.pt)
josé M falçao (marinho.falcao@insa.min-saude.pt)

Version: 2 Date: 25 May 2012

Author's response to reviews:

Methods section - documentation of the permission obtained to use the data from the National Institute of Health.